CeMPA Working Paper Series

CeMPA WP 6/20 Did the UK policy response to Covid-19 protect household incomes?

Mike Brewer Iva Valentinova Tasseva

June 2020



Did the UK Policy Response to Covid-19 Protect Household Incomes?*

Mike Brewer^{1,2} and Iva Valentinova Tasseva¹

¹Institute for Social and Economic Research, University of Essex ²Resolution Foundation

Last version: October 1, 2020

Abstract

We analyse the UK policy response to Covid-19 and its impact on household incomes in the UK in April and May 2020, using microsimulation methods. We estimate that households will lose a substantial share of their net income (7% on average). As a proportion of income, the losses due to the crisis are largest for previously higher-income families. However, the overall impact of the crisis on income inequality is small. Earnings subsidies (the Coronavirus Job Retention Scheme) will protect household finances and provide the main insurance mechanism during the crisis. Besides subsidies, Covid-related increases to state benefits, as well as the automatic stabilisers in the tax and benefit system, will play an important role in mitigating the income losses. Analysing the impact of a near-decade of austerity on the UK safety net, we find that, compared to 2011 policies, the 2020 pre-Covid tax-benefit policies would have been less effective in insuring incomes against the shocks. The extra benefit spending in response to the pandemic will strengthen the safety net, providing important additional income protection.

JEL codes: D31, E24, H24

Keywords: Covid-19; income distribution; earnings subsidies and tax-benefit policies

^{*}Acknowledgements: We are grateful to Holly Sutherland and Tom Lee for insightful comments and suggestions. The results presented here are based on UKMOD version A1.6. UKMOD is maintained, developed and managed by the Institute for Social and Economic Research (ISER) at the University of Essex and we are indebted to Holly Sutherland, Paola De Agostini and the many others who have contributed to its development. The process of extending and updating UKMOD is financially supported by the Nuffield Foundation, based on the annual update of the UK component of EUROMOD funded by the European Union Programme for Employment and Social Innovation "EaSI" (2014-2020). We make use of micro-data from the Family Resources Survey made available by the Department of Work and Pensions and Understanding Society COVID-19 Study by ISER, University of Essex via the UK Data Service. The results and their interpretation are our own responsibility.

1 Introduction

This paper studies the fiscal and distributional impact of Covid-19 in April and May 2020 in the UK. We assess how much income protection is provided by the new earnings subsidies (the Coronavirus Job Retention Scheme and the Self-Employment Income Support Scheme) and Covid-related increases to benefits, as well as looking at how effective are the tax-benefit automatic stabilisers in protecting household finances during recessions. We also analyse how austerity over the last decade has affected the UK safety net.

In more detail, our baseline is the income distribution before Covid-19. Then, using individual-level data from the Understanding Society COVID-19 Study on employment and earnings changes by sex, age and industry, we simulate similarlysized employment and earnings shocks on the sample of workers from the Family Resources Survey 2018/19. Workers affected by the shocks become unemployed or furloughed, or stay employed (not furloughed) but their hours and earnings fall. Furloughed workers receive earnings subsidies from the Coronavirus Job Retention Scheme while self-employed can receive support from the Self-Employment Income Support Scheme. Using the tax-benefit model UKMOD, we then calculate household income tax liabilities and benefit entitlements before and after the employment and earnings shocks. The changes in income taxes and state benefits due to the shocks capture the automatic stabilisation response of the tax-benefit system before the Covid-related benefit increases (Auerbach and Feenberg 2000). Finally, we estimate the effect of Covid-related increases to state benefits on household incomes, keeping fixed the earnings distribution. The analysis provides an average estimate of the distributional impact of Covid-19 during the period of the fullest lockdown, i.e. averaged over the months of April and May 2020.

We find that households will lose a substantial proportion of their income, of 6.9% on average, due to Covid-related employment and earnings shocks. The earnings subsidies for furloughed workers from the Coronavirus Job Retention Scheme will protect household finances from the shocks, accounting for 13.2% of baseline net income. They will support households across the entire income distribution and will provide the main insurance mechanism against the negative income shocks. The Self-Employment Income Support Scheme will also contribute to income gains at all parts of the distribution, of 1.3% on average. Net of the subsidies, the loss in earnings from both employment and self-employment will amount to a substantial 12.6% of net income.

Besides the earnings subsidies, the Covid-related benefit increases and tax-benefit automatic stabilisers will also have an important role in mitigating income losses. Comparing the impact of the benefit increases and automatic stabilisers on net incomes, it is the latter that will have the bigger effect on net incomes, underlining the importance of tax-benefit designs in protecting household incomes during economic downturn. On the other hand, different parts of the distribution rely on different types of policy for mitigating income shocks, stressing the importance of both automatic stabilisers and governments' responses to crises in determining the amount of income protection.

Income from social security benefits – both the system that was in place before the crisis, and the Covid-related increases – contribute to income gains mostly at the bottom of the distribution. Households will benefit mainly from means-tested benefits such as Universal Credit (UC), but the unemployment benefit (Jobseeker's Allowance) will also contribute to small income gains across all parts of the distribution, highlighting the importance of provision and access to social insurance benefits.

Assessing the impact of a near-decade of austerity on the UK safety net, we show that, compared to the 2011 policies, the 2020 *pre-Covid* tax-benefit policies would have been less effective in providing insurance against the shocks for all parts of the distribution. In particular the 2011 system would have raised net income by additional 2% in the poorest decile compared to the 2020 *pre-Covid* system. The extra Covid benefit spending will strengthen the UK safety net, so compared to the 2011 system, the 2020 *post-Covid* system would provide higher levels of income protection in the bottom 5 deciles. But although the Covid-related benefit increases strengthen the safety net, these measures are due to last only during the current financial year.

This paper relates to several strands of literature. It relates to the now fastgrowing literature on the economic and distributional impact of Covid-19. Using the Understanding Society COVID-19 data, Crossley et al. (2020) analyse the economic impact of the pandemic in the UK on employment and earnings, and the ways different household types mitigate the negative shocks, highlighting the importance of transfers from family and friends. Using timely data on the US, Han et al. (2020)assess the impact of the coronavirus crisis and increases to government assistance on US poverty. Adams-Prassl et al. (2020) assess differences in the labour market impact of Covid-19 in the UK, Germany and US. Brewer and Gardiner (forthcoming) analyse the design and beneficiaries of the UK policy response to the pandemic, focusing in particular on low-income households. By simulating employment shocks and using a tax-benefit model, Figari and Fiorio (2020), Beirne et al. (2020) and Bronka et al. (2020) assess the distributional impact of Covid-19 in Italy, Ireland and the UK, respectively. There has been also growing evidence on the impact of the crisis on the US labour market (Coibion et al. 2020b; Cajner et al. 2020) and private spending (Baker et al. 2020; Coibion et al. 2020a).

The current paper builds on and contributes to the existing literature in several ways. First, as survey micro-data on household incomes during the pandemic will only become available with a few years' lag, the present paper addresses this data limitation by combining different data sources and a tax-benefit model to predict household circumstances during Covid-19. Second, the paper brings new insights into the importance of different types of policy in mitigating income losses along the income distribution and quantifies the income compensation provided by the policy response. It also examines empirically how policy reforms over the last decade have reduced the ability of the UK safety net to cushion negative economic shocks. As such, it also contributes to the evidence on the mitigating role of tax-benefit policies during recessions. Income taxes and unemployment insurance benefits have been found to be particularly important in providing income protection and income and consumption smoothing during recessions (Larrimore et al. 2015; Fernández Salgado et al. 2014; Dolls et al. 2012; Kniesner and Ziliak 2002a,b; Auerbach and Feenberg 2000). Furthermore, means-tested benefits have been shown to be effective in providing a much-needed safety net for households during economic downturns (Bitler et al. 2017; Bitler and Hoynes 2016). Recent analysis has also highlighted the importance of tax and benefit automatic stabilisers for income redistribution (Paulus and Tasseva 2020).

The rest of the paper is structured as follows: Section 2 describes the UK policy response to Covid-19 and the tax-benefit policies that will play a role in protecting household incomes from the negative income shocks. Section 3 discusses the decomposition approach to identify the effect of earnings subsidies, tax-benefit automatic stabilisers and Covid-related benefit increases; the data and the simulation of employment and earnings shocks; and the tax-benefit model UKMOD. Section 4 discusses the results and section 5 concludes.

2 The UK policy response to Covid-19 and taxbenefit policies

During the Covid-crisis, access to unemployment and income-related benefits as well as income taxes and national insurance contributions will provide insurance against the economic shocks. In addition, the UK government introduced a package of policy measures in response to Covid-19. These include income protection schemes for workers and increases to state benefits. This section describes the Covid-related policy measures, as well as discussing which tax-benefit policies will play a role in stabilising household incomes during the crisis.

To support businesses and workers, the Coronavirus Job Retention Scheme (hence-

forth CRJS) was introduced to subsidise the earnings of furloughed employees. This allows companies to reduce the hours of workers to zero without laying them off, removing the costs of searching and re-hiring workers later on. In April and May, the CJRS paid 80% of gross earnings up to a maximum of £2,500 per month. Initially the scheme was set for 4 months but on May 12, 2020 it was extended until the end of October. The Office for Budget Responsibility estimates the cost of the scheme to £60 billion for the 8-months duration of the scheme (OBR, 2020). Similarly, the Self-Employment Income Support Scheme (hereafter SEISS) was introduced to provide support for self-employed. The scheme was opened for claims on May 12. It is calculated based on the person's average trading profit over the 3 previous tax years and equals 80% of the average profit, paid in a single instalment covering three-months worth of profit, up to £7,500 in total.¹ OBR estimates the scheme to cost £15 billion (OBR, 2020).

In addition to the earnings subsidies, families affected by the economic shocks can access a range of state benefits to compensate for the losses in earned income. The main ones are the unemployment benefit, Jobseeker's Allowance (JSA), and the means-tested benefit Universal Credit (UC). The contribution-based Jobseeker's Allowance (JSA) is a flat-rate unemployment benefit paid up to 6 months while looking for a job. The benefit is available to those who have paid employees' national insurance contributions. It has two rates: for those aged under 25 ($\pounds 58.9$ per week) and a more generous rate for those aged 25+ (£74.35 per week). UC is an incometested benefit for working-age people on low-incomes or who are unemployed. It was introduced in 2014 with the aim to gradually replace a range of income-related benefits and tax credits, by combining them into a single state transfer. The benefit consists of a basic 'standard allowance' and extra payments which depend on the person's and their family's circumstances, including additional amounts in respect of children, and to those who are renting. UC is withdrawn in line with the joint earnings of the family, and overall entitlement is subject to a maximum ceiling known as a benefit cap.² Prior to Covid-19, the UC rate for singles aged 25+ was the same as JSA, $\pounds 323$ per month, while the rate for couples was about 1.6 times higher, £507 per month.

In response to the crisis, the UK Government increased the level of UC for all family types by £20 a week. In relative terms, this is a substantial increase of 28% for singles aged 25+ and 17% for couples. Access to UC for the self-employed

¹Individuals, whose business has been adversely affected by the pandemic, can apply for the grant. Their profits should not exceed $\pounds 50,000$ per year and must account for more than half of the person's income. Individuals must have submitted Self Assessment tax return for the tax year 2018 to 2019.

 $^{^{2}}$ If due to the change in their circumstances families become subject to the benefit cap, they can be exempt from it by the 39 weeks "grace period". However, the cap will bite if already prior to the shock families were not exempt from it.

was also improved by suspending the Minimum Income Floor: this reduced UC entitlement for those earning less than the equivalent of a certain number of hours at the minimum wage rate. Other Covid-related changes included increases to Housing Benefit (HB) and the UC component which support low-income families with paying their rent. The maximum benefit amount is determined by the so-called Local Housing Allowance (LHA) rates, which have been generally lagging behind price growth since 2012. In response to Covid-19, the LHA rates were increased and realigned to the 30th percentile of the distribution of private market rents. Finally, the amount of earnings disregarded in calculating the entitlements to HB and Council Tax Reduction (a benefit that supports families with paying property tax) was also increased (from £17.1 to £37.1 per week). As with UC, the standard allowance rates for Working Tax Credit – an in-work benefit and one of the benefits that UC is replacing – also went up by £20 a week. At the time of writing, these changes were due to expire in April 2021. OBR estimates the costs for the benefit increases to £8 billion (OBR, 2020).

One contribution of this paper is to identify how the tax and benefit system automatically responds to cushion the impact of labour market shocks. As well as the means-tested benefit system, income tax and national insurance contributions (NIC) also have an important role to play in mitigating income losses. The income tax schedule in the UK is progressive, as are NIC for most parts of the distribution. Thus, as earnings drop due to the economic shocks, income tax payments and NIC also decrease, so that after-tax income drops by less than pre-tax income.

3 Data and Methodology

3.1 Estimating the distributional impact of the crisis

Our aim is to estimate the impact of the crisis on the income distribution using the methodology from Paulus and Tasseva (2020) and Bargain and Callan (2010) who combine household micro-data with tax-benefit microsimulation techniques. We also decompose the changes in the income distribution to identify separately the distributional impact of: earnings losses; earnings subsidies; tax-benefit policies before the Covid-related benefit increases (which we refer to as the automatic stabilisation response of policies); and the Covid benefit increases.

Impact of the crisis

Let us denote with y gross (pre-tax) market income; with t(y) income tax and NIC, which are a function of y; and with b(t, y) state benefits which are a function of t and y. Household net income B (for *baseline*) is then:

$$B = y - t(y) + b(t, y) \tag{1}$$

An economic shock occurs which raises unemployment and lowers gross earnings. The UK Government responded by introducing earnings subsidies and increases to state benefits, so that gross market income after the shock (y_k) plus subsidies (k) is y'; income tax liabilities and NIC after earnings changes are t(y'); and state benefits after earnings and benefit changes are b''(t, y'). Household net income after the shock can be shown to be:

$$D = y' - t(y') + b''(t, y')$$
(2)

and the difference between D and B gives the impact of the crisis on household incomes.

Decomposing changes in the income distribution

We can introduce an intermediate counterfactual scenario C after the shock and earnings subsidies, but before the crisis-related benefit increases. This allows us to decompose the change in income Δ :

$$\Delta = D - B =$$

$$\underbrace{D - C}_{\text{benefit policy changes}} + \underbrace{C - B}_{\text{earnings changes+automatic stabilisers}} =$$

$$\underbrace{y' - t(y') + b''(t, y') - (y' - t(y') + b(t, y'))}_{\text{benefit policy changes}}$$

$$\underbrace{y' - t(y') + b(t, y') - (y - t(y) + b(t, y))}_{\text{earnings changes+automatic stabilisers}}$$
(3)

where the difference between D and C gives the effect of benefit increases (P), keeping fixed the distribution of gross market incomes, tax liabilities and NIC. The difference between C and B gives the total effect of earnings changes plus automatic changes to income tax, NIC and benefits, keeping fixed the tax-benefit rules at their pre-crisis levels (i.e. effect of automatic stabilisers).

Let us now denote as I a functional of income. If I is additively decomposable by income source, e.g. mean net income, we can break down the difference between C and B further into earnings losses (E), earnings subsidies (K), income tax and NIC as automatic stabilisers (S^t) and benefits as automatic stabilisers (S^b) :

$$E = I[y_{k}] - I[y]$$

$$K = I[y'] - I[y_{k}]$$

$$S^{t} = I[t(y)] - I[t(y')]$$

$$S^{b} = I[b(t, y')] - I[b(t, y)]$$
(4)

If I is not additively decomposable by income source, e.g. an inequality indicator such as the Gini coefficient, we approximate the contribution of automatic stabilisers by taking the difference in I based on different income concepts:

$$E + K = I[y'] - I[y]$$

$$S^{t} = I[y' - t'(y)] - I[y - t(y)] - (E + K)$$

$$S^{b} = I[y' - t'(y) + b(t, y')] - I[y - t(y) + b(t, y)] - (I[y' - t'(y)] - I[y - t(y)]) =$$

$$I[C] - I[B] - S^{t} - (E + K)$$
(5)

where I[y'-t'(y)]/I[y-t(y)] is estimated based on gross market income net of taxes and NIC.³

3.2 The data and simulation of shocks

We use the Family Resources Survey (FRS) micro-data for 2018/19 (DWP, 2020), combined with information on employment and earnings shocks from the Understanding Society (UK Household Longitudinal Study-UKHLS) COVID-19 Study (University of Essex, 2020a). The FRS 2018/19 are the latest FRS data available at the time of writing. The FRS can be considered as the benchmark for household incomes data in the UK. They are used for official income statistics by the Department for Work and Pensions, and by the Institute for Fiscal Studies in their annual reports on income trends. We then use information on the labour market situation and earnings of individuals in April and May 2020 from the UKHLS COVID-19 Study to simulate similarly-sized employment and earnings shocks in the FRS.

In more detail: we uprate financial values in the FRS income data to (pre-Covid-19) 2020 prices to account for growth in average earnings and statutory indexation for disability benefits, public pensions, statutory maternity pay, maternity allowance and the statutory sickness pay (we do not directly simulate changes in the labour

³We can rewrite the effect of earnings changes and automatic stabilisaters as: $I[y' - t(y') + b(t,y')] - I[y - t(y) + b(t,y)] = I[y'] - I[t(y')] + I[b(t,y')] - (I[y] - I[t(y)] + I[b(t,y)]) + \eta$, where η is a residual term. For additively decomposable measures, such as mean income, η is zero but it is non-zero for measures which are not additively decomposable by income source, such as Gini. Hence, when we decompose the change in inequality indicators, our estimate for the contribution of automatic stabilisers contains a residual.

market and population structure since 2018-19). These are the data used in the *baseline*, i.e. before the employment and earnings shocks, and so they are our simulation of what the UK population would have looked like in April and May 2020 had there been no coronavirus crisis.

Next, we use the UKHLS COVID-19 Study as our source of data on what sort of employment and earnings shocks have been caused by the crisis. Our aim is to simulate similar-in-size shocks on the FRS sample of workers such that the FRS data with simulated shocks form the underlying data for scenarios C and D.

The UKHLS is a long-running annual household panel study, and the UKHLS COVID-19 data, which were collected through an on-line and phone survey in April and May 2020, were an addition to the usual annual waves that collected information on the labour market situation and earnings of individuals in April and May 2020 and how these had changed since February 2020 (i.e. before the crisis had affected the labour market in the UK). We estimate two multinomial logit models on the UKHLS data, separately for the April and May waves, one on a sample of employed adults and one on a sample of self-employed adults, both restricted to those who had positive earnings in February 2020 (before Covid-19). For the sample of formerly employed workers, the dependent variable is employment status in April/May 2020 and has four outcomes: i) still employed and with no drop in earnings, ii) still employed (not furloughed) but with reduced working hours and earnings, iii) furloughed, and iv) not employed.⁴ The control variables include sex, age and industry (13 categories) and their interactions; household type; baseline earnings quintile; and number of baseline working hours in bands by sex. For the sample of formerly self-employed workers, the dependent variable is self-employment status in April/May 2020 with the following three outcomes: a) still self-employed and with no drop in earnings, b) still self-employed but with reduced hours and earnings, and c) no longer engaged in self-employment. The control variables include sex, age, industry, household type, baseline earnings ventile, and number of baseline working hours in bands by sex. In addition, the May wave of the UKHLS COVID-19 data includes information on entitlement to the SEISS grant. We run a logit model on the sample of formerly self-employed workers to estimate the likelihood of receiving the SEISS grant, controlling for the change in self-employment status since the coronavirus crisis. Results from the multinomial logit, based on the April and May waves separately, are reported in Table A.1 (for the employed) and Table A.2 (for the self-employed), and results from the logit model, based on the May wave only, are reported in Table A.3.⁵

⁴No drop in earnings constitutes a drop of no more than £5 per week. Reduced hours and earnings implies a fall of more than £5 in earnings and at least 1 working hour per week.

⁵Industry was collected at Wave 9 (2017-18) of Understanding Society (University of Essex, 2020b).

We then use the estimated coefficients from the multinomial logit to predict the probability of each outcome for each individual in the FRS sample of workers with positive earnings. We than randomly assign each individual to one of the outcomes, respecting these predicted probabilities. Following the same approach, we use the estimated coefficients from the logit model and the simulated self-employment status after the shock to predict the probability for being entitled to the SEISS grant. In total, 1.1 million workers in the FRS are simulated to be newly unemployed (having applied the grossing weights), 7.3 million workers to be furloughed and 3.1 million workers to experience a drop in working hours and earnings (but not furloughed). Thus, the majority of workers affected by the coronavirus shock will experience a drop in earnings rather than a complete loss in earned income. Overall, 1.8 million self-employed are simulated to receive a grant from the SEISS.

Self-employed predicted to be entitled to the SEISS grant receive the subsidy in May only (when it was first paid out). We estimate the size of the grant as if it was made on a monthly basis (rather than the actual size worth three months of profits) and base our calculations on (baseline) total gross self-employed earnings (FRS variable *seincam2*).⁶ Employees who are predicted to be furloughed (outcome iii) receive the CJRS subsidy in April and May. To calculate how much workers are entitled to under the scheme, we use information on (baseline) total gross earnings from employment (FRS variable *inearns*). Finally, employed and self-employed in the FRS with hours and earnings reduction (outcomes ii and b, respectively) experience a drop in earnings and hours which we simulate as the mean relative drop in earnings by sex in the UKHLS COVID-19 sample amongst those reporting a drop. For employed, the drop is 41% for men and 43% for women in April and 21% for both men and women in May. For self-employed, the drop is 74% for both men and women in April and 56% for women in May.

Table 1 shows the characteristics of workers in the FRS by employment status after the simulated shocks. There is a clear age, education and income gradient among the workers affected by the shocks. A larger share of the newly unemployed are aged 20-29 compared to the other age groups. The newly unemployed are also more likely to come from the lowest earnings quintile. Furloughed workers are more likely to be young, male, lower educated, and to be from the bottom and middle of the earnings distribution. In comparison, those with reduced working hours and earnings (but not furloughed) are more likely to be older, male, and in the bottom earnings quintile. Those with no drop in their earnings ('no change') are more likely to be older, female, higher educated, and from the upper part of the earnings

 $^{^{6}}$ In reality, the grant is calculated based on the average of last three-years profits and provided only to those with profits up to £50,000 per year. As we do not observe this information in the FRS and UKHLS data, we do not account for it in the simulations.

	Unemployed	Furloughed	Reduced hours	No change
			& earnings	
Age group:				
20-29	6.0	34.5	7.1	52.4
30-39	3.9	25.6	10.7	59.7
40-49	3.5	23.4	11.9	61.3
50-59	2.7	22.2	13.5	61.6
60+	3.0	24.6	16.4	56.1
Men	4.0	27.7	12.1	56.2
Women	3.9	24.5	10.2	61.5
In a household with:				
children	4.8	26.7	10.4	58.1
1 earner	3.4	25.4	11.8	59.4
2+ earners	4.1	26.5	11.0	58.5
Completed education aged:				
16 or less	3.8	29.6	12.0	54.6
17-19	3.9	28.6	10.6	56.9
20+	4.1	20.9	11.0	64.0
Earnings quintile:				
1	8.5	31.8	15.2	44.5
2	2.9	33.5	11.9	51.7
3	2.7	29.2	9.9	58.3
4	3.6	22.2	8.7	65.5
5	2.5	15.4	10.8	71.4
Number of (in thousand):				
employed	956	7,291	1,343	14,621
self-employed	140	. 0	1,769	1,722
all	1,096	7,291	3,112	16,343
Receiving SEISS grant	114	0	1.185	491

 Table 1: Characteristics of workers by employment status

Notes: The table shows the share of workers (in %) by different characteristics simulated to be newly unemployed, furloughed, with reduced hours and earnings (not furloughed) and with no drop to earnings (no change). The results are based on average monthly estimates, except for the number of those receiving the SEISS grant which refers to May only. The sample includes individuals with positive earnings from employment and/or self-employment and aged 20-63. Employed with earnings from employment only. Self-employed with earnings from self-employment (and employment).

Source: Own calculations with FRS and Understanding Society COVID-19 data.

Table 2 and Table 3 show the distribution of workers in the FRS by industry and occupation, respectively, by employment status after the simulated shocks. Workers in accommodation and food services, construction and real estate, and the arts, entertainment and recreation are most likely to become unemployed, furloughed or have their earnings and hours reduced, as are workers in elementary, skilled trades and process, plant and machine operatives occupations.

Table 2: Share of workers (in %) by industry

	Unemployed	Furloughed	Reduced hours	No change
			& earnings	
Agriculture, Mining, Manufacture, Utilities	5.1	36.9	7.9	50.1
Construction and Real Estate	2.4	38.4	19.8	39.4
Wholesale and Retail Trade	5.6	36.1	6.8	51.5
Transportation and Storage	1.8	35.1	13.2	49.9
Accommodation and Food	11.5	57.2	10.2	21.1
Information and Communication	3.1	8.5	10.7	77.8
Financial and Insurance Activities	2.2	6.2	4.0	87.7
Professional, Scientific and Technical Activities	1.8	18.4	14.8	65.0
Administrative and Support Service	2.3	31.1	21.6	45.1
Public Administration and Defence	1.8	8.2	4.6	85.4
Education	4.0	13.7	10.6	71.7
Human Health and Social Work	4.5	12.1	9.0	74.4

Arts, Entertainment, Recreation and Other	2.0	37.3	18.8	42.0

Notes: The table shows, for a given sector, the share of workers (in %) simulated to be newly unemployed, furloughed, with reduced hours and earnings (not furloughed) and with no drop to earnings (no change). The results are based on average monthly estimates. The sample includes individuals with positive earnings from employment and/or self-employment and aged 20-63.

Source: Own calculations with the FRS and Understanding Society COVID-19 data.

	Unemployed	Furloughed	Reduced hours	No change
			& earnings	
missing	.0	17.9	6.1	76.0
Managers, Directors and Senior Officials	3.0	24.6	12.3	60.2
Professional Occupations	3.0	15.3	9.7	72.1
Associate Prof. and Technical Occupations	2.7	18.8	11.0	67.5
Admin and Secretarial Occupations	3.3	26.4	7.4	62.9
Skilled Trades Occupations	4.1	32.8	19.5	43.6
Caring, Leisure and Other Service Occupations	4.9	21.3	11.2	62.6
Sales and Customer Service	7.0	36.0	6.0	51.0
Process, Plant and Machine Operatives	4.2	35.5	13.5	46.8
Elementary Occupations	5.9	44.7	10.1	39.2

Table 3: Share of workers (in %) by occupation

Notes: The table shows, for a given occupation, the share of workers (in %) simulated to be newly unemployed, furloughed, with reduced hours and earnings (not furloughed) and with no drop to earnings (no change). The results are based on average monthly estimates. The sample includes individuals with positive earnings from employment and/or self-employment and aged 20-63.

Source: Own calculations with the FRS and Understanding Society COVID-19 data.

3.3 The tax-benefit model UKMOD

The measure of household net income in this paper is cash income and is the sum of gross market incomes, state pensions, national insurance and means-tested benefits minus income tax and NIC liabilities. To account for household composition and economies of scale, household net incomes are equivalised using the modified OECD equivalence scale (with a value of 1 for a couple without children).

To calculate income tax liabilities, NIC and entitlements to benefits, we use the tax-benefit model UKMOD, a spin-off model of the UK component in the EU-wide tax-benefit model EUROMOD (see Sutherland and Figari (2013) and Figari et al. (2015) for information on EUROMOD and Reis and Tasseva (2020) for information on UKMOD). UKMOD simulates for each individual and household in the FRS their taxes and benefits based on their gross earnings, other sources of income, their individual and household characteristics, and the tax-benefit measures that were in place at a given time period. UKMOD calculations account for non-take-up of income-related benefits (see Appendix C).

In the baseline and scenario C, UKMOD calculates net incomes based on the tax-benefit rules that would have applied in April/May 2020 in a world where the Covid-19 did not happen, that is, using the policies that were confirmed in the government Budget on 11 March 2020. The amount of income tax and NIC paid and state benefits received by households differ between the baseline and scenario C because of the simulated shocks in C. By comparing the baseline and scenario C,

we can estimate the fiscal and distributional impact of the earnings subsidies and assess the effectiveness of the pre-Covid tax-benefit system in cushioning economic shocks. We refer to this as the automatic stabilisation effect of policies (see Paulus and Tasseva 2020, Dolls et al. 2012).

In scenario D, UKMOD re-calculates net incomes accounting for the benefit increases that the UK government introduced in response to the crisis. Thus, although the earnings data are the same in scenarios C and D, the amount of state benefits received by households differ because scenario D includes the Covid-related increases in benefit levels (there are no changes in taxes paid by households between scenarios C and D).

UKMOD calculations for scenarios C and D are done separately for April and May. In the remainder of the paper, we show results for the income distribution in scenarios C and D, based on household incomes averaged over April and May.

4 Results

This section assesses the impact of the crisis, and of the UK policy response to Covid-19, on the fiscal budget and the household income distribution. It also looks in more detail at the impact of the increase to state benefits announced by the government when the crisis hit, as well as at how effective the tax-benefit system would have been in responding to the economic shocks without these measures. Finally, it analyses the impact of a near-decade austerity on the UK safety net by assessing how the tax and benefit system of 2011-12 would have protected households from the Covid shock.

4.1 Changes to earnings and the fiscal budget

Table 4 reports the level and changes to total employee and self-employed monthly earnings, subsidies from CJRS and SEISS, government revenues from income tax and NIC, and government spending on state benefits. Column 2 shows the levels in the baseline (before Covid-19). Column 3 shows the impact of the Covid employment and earnings shocks and the full UK policy response (i.e. scenario D versus the baseline) in million £. Table 5 shows the baseline number and the change to the number of earners, taxpayers and state benefit recipients.

Earnings and earnings subsidies from the CJRS and SEISS

The labour market shock means that total employee and self-employed earnings will fall by £8 billion (11.4%) and £3.2 billion (33.4%), respectively. This will be mainly due to workers and self-employed experiencing a cut to their earnings rather than

losing their job. Subsidies from the CJRS will amount to a substantial £10.7 billion, or 15.3% of total employee earnings in the baseline. Spending on grants from the SEISS will amount to £2.1 billion, or 22% of total baseline self-employed earnings. The government will get about 20.1% of the spending on CJRS and SEISS back in the form of income tax and NI contributions.

The shock is simulated to reduce the number of employed and self-employed workers by 956,000 and 140,000, respectively, while 7.3 million workers are simulated to be furloughed and on the CJRS and 1.8 million self-employed to receive a grant from the SEISS. This compares to an estimated 8.7 million jobs that had been furloughed and 2.5 million claims to the SEISS by 31 May.⁷

Income tax and NIC

As earnings fall, revenues from income tax will also fall by a substantial £2.3 billion (16.2%) per month, and revenues from NIC by £1 billion (13%). This translates to 1.5 million people being fully exempt from paying income tax and NIC as their income falls below the Personal Allowance and the NIC Primary Threshold.

Benefit spending

Benefit spending will go up substantially. This will be mainly due to the spending expansion of the main means-tested benefit, Universal Credit (UC), as well as the contributory unemployment benefit, Jobseeker's Allowance (JSA). There will be also increased spending on the Working Tax Credit (WTC), Housing Benefit (HB), Council Tax Reduction (CTR) and Child Benefit (CB).

Spending on UC will expand substantially by £0.8 billion (28.2%) and the number of UC claimants is simulated to rise by 1.1 million (27.2%) to 5.2 million. Thus, UC will provide important safety net during this crisis. For comparison, based on figures published by the Department for Work and Pensions, the record number of claims led to 2.4 million starts to UC in the period 13 March 2020 to 14 May 2020 bringing the total number of people on UC to 5.3 million.⁸ The savings made due to the benefit cap applied on UC are noteworthy, and are simulated to be worth £334 per affected family per month, on average (although they make up a small proportion of the total spend on UC).

Spending on JSA will also increase substantially by $\pounds 261$ million, and the number of JSA claims is simulated to rise by 837,000 due to the sharp rise in the number of unemployed, highlighting the effectiveness of unemployment benefits in providing insurance during economic downturn (Dolls et al. 2012; Auerbach and Feenberg

⁷https://www.gov.uk/government/collections/hmrc-coronavirus-covid-19-statistics

⁸https://www.gov.uk/government/publications/universal-credit-statistics-29-april-2013-to-9-july-2020/universal-credit-statistics-29-april-2013-to-9-july-2020#starts-on-uc-header

2000). Spending on WTC, HB (for non-working age families) and CTR will also rise, providing additional income protection for families. The number of families simulated to be entitled to CB increases, as 159,000 families with a higher earner see their earnings fall sufficiently so that they do not see their CB entitlement withdrawn.

Looking at the combined change in income tax and NI revenues less spending on state benefits, we simulate the shock in April-May to have reduced the net fiscal balance by £16.3 billion per month. About 72% of this will be due to subsidies through CJRS and SEISS, 20% due to losses in income tax and NIC revenues, and 7% due to increased benefit spending.

	Baseline	Impact of crisis
	(in levels)	(change to baseline)
Employee earnings	69,906	-7,961
Self-employed earnings	9,593	-3,208
Earnings subsidies from CJRS	0	10,690
Earnings subsidies from SEISS [*]	0	2,114
Revenues from income $\tan + \text{NIC}$	22,005	-3,317
Income tax	14,308	-2,317
Employee NIC	7,266	-891
Self-employed NIC	432	-109
Spending on benefits	6,163	1,206
Universal Credit (UC)	2,860	805
Savings from UC benefit cap	27	14
Jobseeker's Allowance (JSA)	10	261
Working Tax Credit (WTC)	76	28
Housing Benefit (HB)	707	24
Child Benefit (CB)	961	26
Council Tax Reduction (CTR)	385	26
Other means-tested benefits	1,164	36
Revenues-spending	15,842	-16,270

Table 4: Change in total earnings, revenues and spending (in million £ and per month)

Notes: *Spending on SEISS for May only. All other results are based on average monthly estimates. Other meanstested benefits include the Child Tax Credit, Income support, income-related Employment and Support Allowance, income-based JSA, Pension Credit, Scottish benefits (Sure Start Maternity Grant and Best Start Grant). No simulations to Statutory Sickness Pay.

Source: Own calculations with UKMOD and FRS.

	Baseline	Impact of crisis
	(in levels)	(change to baseline)
Employed	26,714	-956
Self-employed	4,167	-140
Furloughed workers on CJRS	0	7,291
Self-employed on SEISS*	0	1,791
Income $\tan + \operatorname{NIC}$ payers	32,769	-1,497
Income tax	29,925	-1,977
Employee NIC	$23,\!908$	-1,047
Self-employed NIC	3,057	-531
Benefit recipients	13,313	1,471
Universal Credit (UC)	4,071	1,107
Affected by UC benefit cap	82	40
Jobseeker's Allowance (JSA)	34	837
Working Tax Credit (WTC)	298	0
Housing Benefit (HB)	2,001	8
Child Benefit (CB)	$7,\!107$	159
Council Tax Reduction (CTR)	5,270	322
Other means-tested benefits	$3,\!357$	57

Table 5: Change in number of earners, tax payers and state benefit recipients (in
thousand)

Notes: *Number of self-employed on SEISS in May 2020. All other results are based on average monthly estimates. Other means-tested benefits include the Child Tax Credit, Income support, income-related Employment and Support Allowance, income-based JSA, Pension Credit, Scottish benefits (Sure Start Maternity Grant and Best Start Grant). No simulations to Statutory Sickness Pay.

Source: Own calculations with UKMOD and FRS.

4.2 The impact of the crisis on the distribution of income

The distributional impact of the shock to earnings, and of the tax-benefit response

Figure 1 shows the distributional impact of the shock to earnings and the taxbenefit response by showing the difference in mean equivalised household net income between the baseline and scenario D, where households are ranked based on their net income before Covid-19, for each decile group (1-10), and for the whole population (All). As well as showing the change in net income (black circle), we also show the change due to employee earnings and self-employed earnings; subsidies from CJRS (where someone is put on the CJRS, this is treated as if they lost all their earnings and then gained a new source of income, CJRS payments) and SEISS; income tax + NIC; and all benefits. A negative change means a fall in net income.

In summary, households will lose a large proportion of their income due to the pandemic, with losses higher in the middle and top of the distribution than at the bottom. Subsidies from CJRS will play a crucial role in supporting household finances across the entire distribution. They will provide the main insurance mechanism during the crisis. Subsidies from SEISS will also provide income support across the distribution, with somewhat larger benefits at the bottom. Reductions in income taxes and NIC will provide additional income protection for households at the top, while increases to benefit entitlements will protect households at the bottom of the distribution.

In more detail, we estimate that the average loss in net income will be 6.9%. Breaking down the change by income source shows that earnings subsidies from the CJRS will account for a large share of the average baseline income, i.e. 13.2%. Subsidies from the SEISS will contribute to a smaller income gain of 1.3% on average. Net of subsidies, the loss in earnings from both employment and self-employment will amount to a substantial 12.6% of mean net baseline income. Automatic reductions in income taxes and NIC will also help mitigate income losses, contributing to an average gain in net income of 4.1%. A smaller gain of 1.6% will come from state benefits.

Across the income distribution, the top eight decile groups will lose out, on average, with losses proportionally larger in the middle and top of the distribution than at the bottom. The fact that the losses in earned income are skewed towards high-income families mostly reflects the way that earners are distributed across the household income distribution: there are substantially more no-earner and fewer two-earner households at the bottom than in the middle or top of the distribution (see Table A.5). If we only focus on households with one earner, then the losses in earned income, as a proportion of baseline net income, are largest at the bottom of the distribution (see Figure B.1). The losses in earned income for households with two-and-more earners are somewhat more equally distributed (see Figure B.2).⁹

The CJRS and SEISS earnings subsidies combined provide a very important safety net across all deciles, amounting to 10.5% of the baseline net income in the poorest decile up to 19.5% in the seventh decile. In the richest decile, the cap on the CJRS subsidy will make the scheme less effective in insuring household incomes against the shocks.

After the earnings subsidies, reductions in income tax and NIC will also help mitigate income losses. They will absorb a higher proportion of the losses the higher the decile group, due to the progressive nature of the income tax and NIC schedule. In other words, income tax and NIC will shield higher-earners more than lower-earners (in fact, in the richest 10th decile, they will mitigate an equal share of the income loss as the CJRS and SEISS subsidies combined).

At the bottom of the distribution, state benefits will play an important role in alleviating pressure on the households budgets. In the bottom three deciles, state benefits will contribute to sizeable income gains of 4-5%.

⁹These results are broadly consistent with the estimates for the distributional impact of the pandemic on working households by HM Treasury (2020).



Figure 1: Impact of the crisis and policy response to Covid-19 on mean net income by decile of *pre-crisis* income

Notes: The figure shows the distributional impact of the employment and earnings shocks and the UK policy response, i.e. the *baseline* versus scenario *D*. Results based on average monthly estimates. Changes in income based on equivalised household net income. *Source:* Own calculations using UKMOD and FRS.

Changes in the distribution of income

Figure 2 presents these results in a different way, by comparing the level and composition of household net incomes in the baseline and scenario D, and doing this within each decile group having re-ranked households. The pattern of income changes is somewhat different from that shown in Figure 1 because the re-ranking effect is so large. Figure 2 shows that the crisis is simulated to lead to a large rise in the number of adults with very low disposable income (caused by those who lose all their earnings as a result of the crisis), such that the bottom decile will see a drop of 3.9%(larger than the drop in deciles two and three). The pattern of changes in income across the decile groups is broadly consistent with a reduction in inequality.

Figure 3 shows the impact of Covid-19 on income inequality based on the Gini coefficient, coefficient of variation (CV), mean log deviation (MLD) and the Theil index (TI). The total change in inequality (black dot) is decomposed into the contri-

bution of earnings changes (i.e. earnings losses net of subsidies) and of tax-benefit policies. In line with Figure 2, we find that net income inequality will fall slightly. Breaking down the change in net income inequality shows that earnings changes and tax-benefit policies will have opposing effects on inequality. Disparities in earnings will increase although the increase will be statistically significant only for the Gini coefficient (0.005). However, tax-benefit policies will more than offset the increased inequality in earned income. We return to the distributional impact of tax-benefit policies in the next section.

Looking at changes in income poverty, we estimate that the poverty rate and poverty gap will increase due to Covid-19 using a fixed poverty line at 60% of the median baseline income, but will fall using a floating poverty line due to a significant drop in median income (see Table A.5 and Table A.6, respectively). For the overall population, the poverty rate will increase from 16.5% to 17.7% with a fixed poverty line while it will go down to 13.7% with a floating poverty line.





Notes: The figure shows the distributional impact of the employment and earnings shocks and the UK policy response, i.e. the *baseline* versus scenario *D*. Results based on average monthly estimates. Changes in income based on equivalised household net income. Households ranking based on the respective (baseline or post-Covid) income distribution.

Source: Own calculations using UKMOD and FRS.





Notes: The figure shows the impact on inequality of the employment and earnings shocks and the UK policy response, i.e. the *baseline* versus scenario *D*. CV=coefficient of variation; MLD=mean log deviation; TI=Theil index. Results based on average monthly estimates of income. Changes in income inequality based on equivalised household net income.

Source: Own calculations using UKMOD and FRS.

4.3 The distributional impact of the Covid-19 increases to state benefits and tax-benefit stabilisers

In this section, we analyse separately the contribution of the state benefit increases announced in response to Covid-19 and the contribution of the tax-benefit system that we would have had in their absence (i.e. the one confirmed in the government's March 2020 Budget): we refer to this latter one as the automatic stabilisers.

Figure 4 repeats the change in net income due to state benefit entitlements, and income tax and NIC liabilities that was shown in Figure 1, but now shows in the left plot, the impact of the automatic stabilisers and in the right plot, the impact of Covid-related benefit increases (for the estimation of the effect of benefit increases and automatic stabilisers, see equations 3 and 4, respectively). Comparing the total impact of the automatic stabilisers versus the total effect of Covid-related benefit increases on mean net incomes, it is clear that the former has the most impact on net incomes, on average. This underlines the importance of tax-benefit designs in protecting household incomes during economic downturn. On the other hand, different parts of the distribution rely on different types of policy for mitigating income shocks, stressing the importance of both automatic stabilisers and governments' responses to crises for providing income protection (see e.g. Paulus and Tasseva 2020).

In more detail, looking at the plot on the left, income taxes and NIC will have the most sizeable effect on mean net income. Furthermore, they compensate most for income losses at the upper end of the distribution, due to the progressivity of the system. In the richest top decile, income tax and NIC will contribute to a gain of 6.3% and 1.3%, respectively. In other words, after the drop in earnings, the top decile will retain in total 7.7% of its baseline income due to lower tax liabilities and NIC. This emphasises the role of income taxes and NIC as an important source for income insurance (Kniesner and Ziliak 2002a,b).

However, income taxes and NIC will have less impact at the bottom of the distribution where, on average, they account for a much smaller share of baseline income (for example, someone earning £16,800 per year, i.e. working 37 hours a week at the National Living Wage level, would pay 5% of gross income in income tax, compared to 15% if earning £50,000 per year).

What protects most households at the bottom of the distribution during recessions are state benefits: in the case of the current crisis it is mainly the means-tested benefit UC. As a stabiliser (left plot), UC will provide a safety-net mainly for the bottom six deciles. The Covid-related increases to UC (right plot) contribute to additional income gains targeted at the bottom of the distribution. For the poorest decile, this would amount to an average gain in equivalised net income of 2.8%. The stabilisation effect of UC and the UC stimulus package will benefit most households with children, lone mothers, one-earner families and those in privately rented or social housing (Figure B.6). After UC, the unemployment benefit JSA (left plot) will also contribute to small income gains across all parts of the distribution highlighting the importance of provision and access to social insurance benefits. For households with two-and-more earners the gains from JSA will be quite sizeable, especially in the bottom four deciles (Figure B.4). The increases to WTC and HB (right plot) will have a small positive effect on net incomes.



Figure 4: Impact of the crisis and policy response to Covid-19 on mean net income by decile and tax-benefit policy

Notes: Left plot shows impact of automatic stabilisers, i.e. the baseline versus scenario C. Right plot shows impact of Covid-related benefit increases, i.e. scenario C versus D. Results based on average monthly estimates. Changes in total net income and the contribution of earnings changes, CJRS subsidies and SEISS grant are omitted. Changes in income based on equivalised household net income. Other benefits include the Council Tax Reduction, Child Tax Credit, Income support, income-related Employment and Support Allowance, income-based JSA, Pension Credit, Scottish benefits (Sure Start Maternity Grant and Best Start Grant). No simulations to Statutory Sickness Pay. Source: Own calculations using UKMOD and FRS.

Next, we look at the impact of tax-benefit policies on income inequality measured by Gini, CV, MLD and TI. Figure 5 shows the separate contribution of benefits (breaking down their contribution into stabilisers and Covid-related benefit increases) and tax+NIC stabilisers to the total change in inequality. The total change in inequality and contribution of earnings changes are omitted. Table A.7 includes all estimates for the baseline level and changes in inequality. The key result is that benefits (both automatic stabilisers and Covid-related benefit increases) will contribute to an inequality reduction, highlighting their importance for redistribution; while income tax and NIC will have little impact on inequality. Overall, benefits will reduce the Gini coefficient by 0.018, CV by 0.036, MLD by 0.016 and TI by 0.017.



Figure 5: Decomposing the change in income inequality by tax-benefit policy

Notes: CV=coefficient of variation; MLD=mean log deviation; TI=Theil index. Results based on average monthly estimates of income. Changes in income inequality based on equivalised household net income. Total change in inequality and contribution of earnings changes are omitted. *Source:* Own calculations using UKMOD and FRS.

4.4 The impact of austerity on the UK safety net

In this section, we explore the impact of a near-decade of austerity on the effectiveness of tax-benefit policies in providing income protection. In the last decade, there have been various cuts made to eligibility, or freezes made to the nominal entitlements (De Agostini et al. 2018; Bourquin et al. 2019). On the tax side, the top marginal tax rate was reduced from 50% to 45% and the zero tax band (Personal Allowance) has grown faster than prices, but the higher income tax thresholds and the point at which the Personal Allowance is tapered away have deteriorated in real terms.

To assess the impact of austerity on the UK safety net, we compare the distributional impact of tax and benefit policies had the 2011 system been in place today (indexed by CPI) versus the 2020 *pre-Covid* and the 2020 *post-Covid* systems (Figure 6). To estimate the impact of the 2011 policies, we use the tax-benefit model UKMOD to apply the indexed 2011 tax-benefit policies on the household micro-data a) before Covid-19 and b) after the simulation of employment and earnings shocks.

Comparing the distributional impact of the 2011 with the 2020 *pre-Covid* system shows that, due to austerity measures, for all parts of the distribution the 2020 *pre-*

Covid system would have been less effective in providing insurance against shocks. In particular in the poorest decile, the 2011 tax-benefit policies would have raised net income by an additional 2.1% compared to the 2020 *pre-Covid* policies.

The emergency benefit package in response to the pandemic will strengthen the UK safety net and, compared to 2011, the 2020 *post-Covid* system will provide more insurance in deciles 1 to 5 and about the same in the rest of the distribution. Nevertheless, although the Covid-related benefit increases provide additional income insurance, these protections are temporary measures and do not undo the impact of austerity measures over the years.





Notes: 2020 pre-Covid and 2020 post-Covid systems based on scenarios C and D, respectively. 2011 tax-benefit parameters in 2020 prices and applied on the earnings distribution before and after Covid-19 to estimate effect of automatic stabilisers.

Source: Own calculations using UKMOD and FRS.

5 Conclusions

Governments across the globe have taken drastic measures to address the economic costs of Covid-19. The UK government introduced income protection in the form of earnings subsidies for workers, and raised the level of means-tested state benefits. Families experiencing income losses due to Covid-19 can rely on state support through these measures as well as through the built-in automatic stabilisation response of the tax-benefit system.

Combining different sources of household micro-data and a tax-benefit model, we estimate the fiscal and distributional impact of Covid-19 during April and May 2020. We assess separately the impact of earnings losses, government's earnings subsidies (the Coronavirus Job Retention Scheme and the Self-Employment Income Support Scheme), state benefit increases and tax-benefit automatic stabilisers.

We simulate that UK households will, on average, sustain income losses in net income due to Covid-19 of 6.9%. Earned income falls by a substantial 12.6% of baseline net income. Earnings subsidies for furloughed workers will play a major role in protecting household incomes, accounting for 13.2% of baseline net income. They will support households across the entire income distribution and will provide the main insurance mechanism against the negative income shocks. The SEISS grant will also provide small income gains of 1.3% on average.

Besides the earnings subsidies, tax-benefit policies will provide needed income protection for families. At the bottom of the distribution, UC as a stabiliser will help families cope with the shocks. Covid-related increases to UC payments will be especially targeted at the poorest decile, contributing to a net income gain of 3%. The unemployment benefit JSA will also help mitigate income losses, providing insurance against the shocks across the entire distribution. But the tax system itself also cushions the shock, and automatic reductions in income tax and national insurance contributions are worth 4.1%, on average, of the pre-Covid baseline income, with the middle and top of the distribution benefiting mostly from income tax and NIC stabilisers due to the progressivity of the tax schedule and NIC.

Finally, we look at the impact of a near-decade of austerity measures on the UK safety net. Compared to the 2011 system, we show that the 2020 *pre-Covid* tax-benefit system would have been less effective in providing insurance against the shocks. The extra Covid benefit spending will strengthen the safety net, so the 2020 *post-Covid* system would provide higher level of income protection at the bottom half of the distribution than the 2011 system. But although the Covid-related benefit increases strengthen the safety net, these protections are temporary measures and do not undo the impact of austerity over the years.

References

Adams-Prassl, A., Boneva, T., Golin, M., and Rauh, C. (2020). "Inequality in the Impact of the Coronavirus Shock: Evidence from Real Time Surveys." Cambridge-INET Working Paper Series No 2020/18, Cambridge: University of Cambridge.

- Auerbach, A. J., and Feenberg, D. (2000). "The Significance of Federal Taxes as Automatic Stabilizers." Journal of Economic Perspectives, 14(3), 37–56.
- Baker, S. R., Farrokhnia, R., Meyer, S., Pagel, M., and Yannelis, C. (2020). "How Does Household Spending Respond to an Epidemic? Consumption During the 2020 COVID-19 Pandemic." Working Paper 26949, National Bureau of Economic Research.
- Bargain, O., and Callan, T. (2010). "Analysing the Effects of Tax-benefit Reforms on Income Distribution: A Decomposition Approach." *Journal of Economic In*equality, 8(1), 1–21.
- Beirne, K., Doorley, K., Regan, M., Roantree, B., and Tuda, D. (2020). "The Potential Costs and Distributional Effect of Covid-19 related Unemployment in Ireland." EUROMOD Working Paper EM05/20, Colchester: University of Essex.
- Bitler, M., and Hoynes, H. (2016). "The More Things Change, the More They Stay the Same? The Safety Net and Poverty in the Great Recession." *Journal of Labor Economics*, 34 (S1), 403–444.
- Bitler, M., Hoynes, H., and Kuka, E. (2017). "Child Poverty, the Great Recession, and the Social Safety Net in the United States." *Journal of Policy Analysis and Management*, 36(2), 358–389.
- Bourquin, P., Cribb, J., Waters, T., and Xu, X. (2019). "Living Standards, Poverty and Inequality in the UK: 2019." Report no. R157, London: Institute for Fiscal Studies.
- Brewer, M., and Gardiner, L. (forthcoming). "The Impact of COVID-19 on Low-Income Households." Oxford Review of Economic Policy.
- Bronka, P., Collado, D., and Richiardi, M. (2020). "The Covid-19 Crisis Response Helps the Poor: The Distributional and Budgetary Consequences of the UK Lock-Down." EUROMOD Working Paper EM11/20, Colchester: University of Essex.
- Cajner, T., Crane, L. D., Decker, R. A., Grigsby, J., Hamins-Puertolas, A., Hurst, E., Kurz, C., and Yildirmaz, A. (2020). "The U.S. Labor Market During the Beginning of the Pandemic Recession." Becker Friendman Institute Working Paper 2020-58.
- Coibion, O., Gorodnichenko, Y., and Weber, M. (2020a). "Labor Markets During the COVID-19 Crisis: A Preliminary View." Working Paper 27141, Becker Friedman Institute Working Paper 2020-41.

- Coibion, O., Gorodnichenko, Y., and Weber, M. (2020b). "The Cost of the Covid-19 Crisis: Lockdowns, Macroeconomic Expectations, and Consumer Spending." Working Paper 27141, National Bureau of Economic Research.
- Crossley, T. F., Fisher, P., and Low, H. (2020). "The Heterogeneous and Regressive Consequences of COVID-19: Evidence from High Quality Panel Data." Department of Economics Discussion Paper Series 919, Oxford: University of Oxford.
- De Agostini, P., Hills, J., and Sutherland, H. (2018). "Were We Really All in it Together? The Distributional Effects of the 2010–15 UK Coalition Government's Tax-benefit Policy Changes." Social Policy & Administration, 52(5), 929–949.
- Department for Work and Pensions, Office for National Statistics, NatCen Social Research (2020). "Family Resources Survey, 2018–2019." [data collection]. UK Data Service. SN: 8633.
- Dolls, M., Fuest, C., and Peichl, A. (2012). "Automatic Stabilizers and Economic Crisis: US vs. Europe." Journal of Public Economics, 96(3–4), 279–294.
- Fernández Salgado, M., Figari, F., Sutherland, H., and Tumino, A. (2014). "Welfare compensation for unemployment in the Great Recession." *Review of Income and Wealth*, 60, 177–204.
- Figari, F., and Fiorio, C. V. (2020). "Welfare Resilience in the Immediate Aftermath of the COVID-19 Outbreak in Italy." EUROMOD Working Paper EM06/20, Colchester: University of Essex.
- Figari, F., Paulus, A., and Sutherland, H. (2015). "Microsimulation and Policy Analysis." In A. B. Atkinson, and F. Bourguignon (Eds.), *Handbook of Income Distribution*, vol. 2B, 2141–2221, Elsevier BV.
- Han, J., Meyer, B. D., and Sullivan, J. X. (2020). "Income and Poverty in the COVID-19 Pandemic." NBER Working Paper 27729, National Bureau of Economic Research.
- HM Treasury (2020). "Impact of COVID-19 on Working Household Incomes: Distributional Analysis as of May 2020." Tech. rep., HM Treasury.
- Kniesner, T. J., and Ziliak, J. P. (2002a). "Explicit Versus Implicit Income Insurance." The Journal of Risk and Uncertainty, 25(1), 5–20.
- Kniesner, T. J., and Ziliak, J. P. (2002b). "Tax Reform and Automatic Stabilization." American Economic Review, 92(3), 590–612.

- Larrimore, J., Burkhauser, R. V., and Armour, P. (2015). "Accounting for Income Changes Over the Great Recession Relative to Previous Recessions: The Impact of Taxes and Transfers." *National Tax Journal*, 68(2), 281–318.
- OBR (2020). "Coronavirus Policy Monitoring Database: 4 June 2020." Tech. rep., Office for Budget Responsibility.
- Paulus, A., and Tasseva, I. (2020). "Europe through the Crisis: Discretionary Policy Changes and Automatic Stabilisers." Oxford Bulletin of Economics and Statistics, 82, 864–888.
- Reis, S., and Tasseva, I. (2020). "UKMOD/EUROMOD Country Report: United Kingdom (UK) 2018–2024." CeMPA Working Paper 7/20, Colchester: University of Essex.
- Sutherland, H., and Figari, F. (2013). "EUROMOD: the European Union Taxbenefit Microsimulation Model." International Journal of Microsimulation, 6(1), 4–26.
- University of Essex, Institute for Social and Economic Research (2020a). "Understanding Society: COVID-19 Study, 2020." [data collection]. UK Data Service. SN: 8644.
- University of Essex, Institute for Social and Economic Research (2020b). "Understanding Society: Waves 1-9, 2009-2018 and Harmonised BHPS: Waves 1-18, 1991-2009." [data collection]. 12th Edition. UK Data Service. SN: 6614.

Supplementary Materials

A Tables

Table A.1:	Multinomial logit of employment status in Ap	oril 2020
	(base outcome = no drop in earnings)	

	Reduced hours	Furloughed	Unemployed
	& earnings		
Constant	-1.948	120	-6.422*
	(2.722)	(.871)	(3.320)
Men	422	633	9.506^{**}
	(2.895)	(1.360)	(4.312)
Age	.005	.014	.089
	(.059)	(.018)	(.058)
Men x Age	.016	007	175**
	(.061)	(.025)	(.081)
Household type:		. ,	

Single adult, no children	.368	092	.029
Single adult living with children	(.376) .590	(.187) 193	(.554) 204
Multiple adults, no children	(.448) 340	(.190) - 002	(.640)
Multiple adults, no children	(.398)	(.192)	(.546)
Multiple adults living with children	.071 (.422)	.149 (.196)	.124 (.658)
Earnings quintile:	()	()	()
1 2	ref 159	ref 313*	ref -2.136***
	(.248)	(.173)	(.613)
3	806** (338)	716*** (184)	-1.171^{*}
4	867**	-1.230***	627
5	(.377) 497	(.196) -1.786***	(.595) -1.788***
	(.385)	(.219)	(.637)
Sex x Hours in work: Women x less than 20	ref	ref	ref
Women x 20-34	692***	187	747*
Mon y 20.34	(.264)	(.203)	(.447)
Men X 20-04	(.828)	(.609)	(.840)
Women x 35+	934***	349	904
Men x $35+$	(.343) - 754	(.234) 1.264^{**}	(.632) 510
	(.616)	(.567)	(1.028)
Sector:	nof	nof	nof
Construction and Real Estate	-8.161**	421	2.081
	(4.098)	(1.449)	(3.484)
Wholesale and Retail Trade	-2.520	838	7.298**
Transportation and Storage	(2.892) 1.923	(1.000) 1.647	(3.470) 453
	(5.230)	(1.853)	(3.972)
Accommodation and Food	163	.473	6.314^{*} (3.584)
Information and Communication	4.024	950	5.539
Financial and Insurance Activities	(3.333) - 800	(2.355)	(3.895) 15.868***
	(3.037)	(1.934)	(4.756)
Professional, Scientific and Technical Activities	.894	.332	8.801**
Administrative and Support Service	-4.514	.142	(4.104) -10.077***
	(3.311)	(1.488)	(3.376)
Public Administration and Defence	-1.383 (4.043)	922 (1.646)	1.699 (6.075)
Education	1.012	800	4.981
	(2.703)	(1.002)	(3.697)
Human Health and Social Work	(2.817)	(1.014)	(3.534)
Arts, Entertainment, Recreation and Other	3.427	3.496***	7.721**
Sex x Sector:	(3.323)	(1.232)	(3.713)
Women x Agriculture, Mining, Manufacture, Utilities	ref	ref	ref
Men x Construction and Real Estate	12.097^{***}	3.530	-8.100^{*}
Men x Wholesale and Retail Trade	(4.455) 2.367	(2.176) 1.445	(4.911) -9.201*
Men y Transportation and Storage	(3.232)	(1.512) 1 120	(4.759)
Men x Transportation and Storage	(5.613)	(2.257)	(5.416)
Men x Accommodation and Food	2.932	3.810**	-8.809*
Men x Information and Communication	(3.456) -5.921	(1.821) -1.330	(5.315) -13.164***
	(4.304)	(3.211)	(4.659)
Men x Financial and Insurance Activities	.145 (4 518)	097 (2.809)	-32.385*** (11 899)
Men x Professional, Scientific and Technical Activities	1.898	-1.881	-22.689***
Men y Administrative and Support Service	(3.959) 7 266**	(1.912)	(5.302)
wen a rummistrative alle support service	(3.676)	(2.264)	(4.447)
Men x Public Administration and Defence	3.099	.603	-9.862

	(5.021)	(2.658)	(7.007)
Men x Education	1.046	-1.151	-9.151*
	(3.104)	(1.972)	(4.738)
Men x Human Health and Social Work	691	.073	19.547
	(3.337)	(1.913)	(15.028)
Men x Arts, Entertainment, Recreation and Other	-1.776	-2.179	-26.624^{***}
	(3.707)	(1.862)	(4.592)
Age x Sector:			_
Age x Agriculture, Mining, Manufacture, Utilities	ref	ref	ref
Age x Construction and Real Estate	.143*	.002	046
	(.080)	(.032)	(.061)
Age x Wholesale and Retail Trade	.043	.011	139**
	(.063)	(.023)	(.066)
Age x Transportation and Storage	033	047	.001
	(.111)	(.041)	(.073)
Age x Accommodation and Food	.023	.010	098
	(.064)	(.026)	(.068)
Age x Information and Communication	072	.014	072
	(.077)	(.051)	(.072)
Age x Financial and Insurance Activities	.007	024	494***
	(.068)	(.042)	(.134)
Age x Professional, Scientific and Technical Activities	030	013	207**
	(.070)	(.026)	(.091)
Age x Administrative and Support Service	.083	005	080
	(.070)	(.032)	(.062)
Age x Public Administration and Defence	.018	029	039
	(.084)	(.035)	(.115)
Age x Education	030	005	091
0	(.059)	(.021)	(.068)
Age x Human Health and Social Work	.000	035	025
5	(.061)	(.022)	(.065)
Age x Arts, Entertainment, Recreation and Other	062	063**	140**
	(.073)	(.026)	(.068)
Sex x Age x Sector:		()	()
Women x Age x Agriculture, Mining, Manufacture, U	ref	ref	ref
Men x Age x Construction and Real Estate	222**	054	.169*
	(.087)	(.046)	(.093)
Men x Age x Wholesale and Retail Trade	058	030	.159*
	(.070)	(.033)	(.096)
Men x Age x Transportation and Storage	- 004	- 016	096
won x 11ge x fransportation and storage	(119)	(048)	(105)
Men x Age x Accommodation and Food	- 040	- 075*	(.100)
won's rige's recommodation and rood	(0.076)	(041)	(119)
Men x Age x Information and Communication	(.010)	- 018	(.110)
Wen x Age x mormation and communication	(096)	(068)	(000)
Mon y Age y Financial and Insurance Activities	(.030)	(.008)	703***
Well x Age x Finalicial and Insulance Activities	(000)	(061)	(230)
Mon y Are y Professional Scientific and Technical Activities	(.033)	(.001)	(.203)
Men x Age x Froiessional, Scientific and Technical Activities	037	.029	.445
Man A no A durinistruction and Commant Compile	(.087)	(.040)	(.117)
Men x Age x Administrative and Support Service	11(003	.191
	(.082)	(.051)	(.089)
Men x Age x Public Administration and Defence	083	.004	.183
	(.106)	(.059)	(.134)
Men x Age x Education	036	.010.	.151
	(.067)	(.039)	(.093)
Men x Age x Human Health and Social Work	017	024	767
	(.071)	(.042)	(.545)
Men x Age x Arts, Entertainment, Recreation and Other	.001	.019	.209**
	(.077)	(.039)	(.090)
N	6122		

(table continued) Multinomial logit of employment status in \underline{May} 2020 (base outcome = no drop in earnings)

	Reduced hours	Furloughed	Unemployed
	& earnings		
Constant	-2.491	181	-13.032***
	(1.549)	(1.096)	(2.312)
Men	1.023	-1.415	12.855***
	(2.037)	(1.702)	(3.281)

Age	.008	.014	.197***
	(.033)	(.022)	(.040)
Men x Age	.011	007	236***
Household type:	(.041)	(.030)	(.056)
Single adult no children	- 091	591**	598
	(.279)	(.210)	(.498)
Single adult living with children	.262	.445**	-1.134*
0 0	(.248)	(.214)	(.617)
Multiple adults, no children	073	.512**	.283
	(.297)	(.234)	(.521)
Multiple adults living with children	252	.588**	1.040
Farnings quintile	(.320)	(.240)	(.059)
1	ref	ref	ref
2	.335	137	542
	(.263)	(.232)	(.370)
3	079	407	496
	(.323)	(.253)	(.489)
4	507	625**	503
5	(.331)	(.257) 1 020***	(.525)
0	(309)	(294)	(501)
Sex x Hours in work:	(.555)	(.201)	(.001)
Women x less than 20	\mathbf{ref}	ref	ref
Women x 20-34	.241	249	-1.409***
	(.315)	(.246)	(.378)
Men x 20-34	-1.650*	1.566*	.723
Warran - 25	(.842)	(.847)	(1.286)
women x 55+	(367)	497	-2.203 (517)
Men x 35+	-1.085*	1.460*	553
	(.615)	(.826)	(1.275)
Sector:			, , , , , , , , , , , , , , , , , , ,
Agriculture, Mining, Manufacture, Utilities	ref	ref	ref
Construction and Real Estate	-7.217**	-1.819	9.513***
Wholesale and Betail Trade	(3.239) 2.437	(1.721) 1 500	(2.610) 13 781***
Wholesale and Retail Trade	(1.962)	(1.225)	(2.402)
Transportation and Storage	-21.692***	.448	8.226
	(6.438)	(1.973)	(8.321)
Accommodation and Food	-1.905	.090	12.381***
	(1.923)	(1.390)	(2.722)
Information and Communication	-2.643	-4.255**	9.428**
Financial and Insurance Activities	(2.061) 1 243	(2.143)	(4.595) 19 646***
Financial and Insurance Activities	(1.982)	(2.419)	(3.626)
Professional, Scientific and Technical Activities	1.923	.608	13.689***
	(2.033)	(1.458)	(2.638)
Administrative and Support Service	.934	.378	-2.475
	(2.477)	(1.558)	(2.439)
Public Administration and Defence	522	-3.334^{*}	9.327*
Education	(2.311) 346	(1.750)	(5.122) 13 103***
Education	(1.658)	(1.301)	(2.873)
Human Health and Social Work	-2.295	453	12.241***
	(2.002)	(1.291)	(2.257)
Arts, Entertainment, Recreation and Other	-1.981	2.666^{*}	15.538^{***}
· · ·	(3.527)	(1.517)	(2.873)
Sex x Sector:	naf	nof	f
Men x Construction and Real Estate	9 346**	3 622	-16 567***
Well x construction and rear Estate	(3.646)	(2.312)	(3.628)
Men x Wholesale and Retail Trade	.791	1.342	-16.268***
	(2.699)	(1.857)	(3.569)
Men x Transportation and Storage	21.571^{***}	.722	-17.409*
	(7.027)	(2.470)	(9.926)
wen x Accommodation and Food	-16.206*** (2 528)	(2.187)	-10.007**
Men x Information and Communication	(2.020) 1 709	2.130 <i>)</i> 2.583	-6.475
	(2.846)	(3.281)	(6.260)
Men x Financial and Insurance Activities	-7.973***	-1.973	-18.664***
	(2.803)	(3.004)	(6.967)

Men x Professional, Scientific and Technical Activities	-1.767	-3.172	-24.255***
	(2.674)	(2.124)	(3.634)
Men x Administrative and Support Service	2.976	4.560^{*}	1.044
Men x Public Administration and Defence	(3.234) 2.249	(2.490) 1 392	(3.729) -13 955**
Wen x I ubic Administration and Defence	(2.986)	(3.527)	(5.902)
Men x Education	.439	161	-21.322***
	(2.358)	(2.318)	(6.169)
Men x Human Health and Social Work	.056	-2.992	-6.418
	(4.075)	(2.476)	(5.239)
Men x Arts, Entertainment, Recreation and Other	-16.722	-3.101	-33.84(****
Age x Sector:	(13.955)	(2.393)	(3.092)
Age x Agriculture, Mining, Manufacture, Utilities	ref	ref	ref
Age x Construction and Real Estate	.135**	.024	165***
	(.062)	(.036)	(.046)
Age x Wholesale and Retail Trade	.035	.019	235***
Anon Transportation and Stanona	(.042)	(.027)	(.047)
Age x Transportation and Storage	.395 (118)	(0.43)	119
Age x Accommodation and Food	.058	(.043)	177***
	(.043)	(.032)	(.052)
Age x Information and Communication	.028	.070	121
	(.042)	(.047)	(.091)
Age x Financial and Insurance Activities	.018	047	240***
	(.043)	(.059)	(.084)
Age x Professional, Scientific and Technical Activities	046	028	250***
Are y Administrative and Support Sources	(.044)	(.031)	(.050)
Age x Administrative and Support Service	(052)	(0.35)	205
Age x Public Administration and Defence	012	.009	164*
	(.050)	(.036)	(.097)
Age x Education	004	011	225***
	(.036)	(.027)	(.053)
Age x Human Health and Social Work	.045	032	210***
	(.044)	(.027)	(.042)
Age x Arts, Entertainment, Recreation and Other	.021	056*	294***
Ser r Age r Sector	(.009)	(.052)	(.004)
Women x Age x Agriculture, Mining, Manufacture, U	ref	ref	ref
Men x Age x Construction and Real Estate	185***	056	.276***
5	(.070)	(.048)	(.062)
Men x Age x Wholesale and Retail Trade	017	024	.271***
	(.059)	(.040)	(.076)
Men x Age x Transportation and Storage	390***	008	.280
Man - Ano - Accommodation and Food	(.131)	(.051)	(.184)
Men x Age x Accommodation and Food	023	021	(083)
Men x Age x Information and Communication	031	083	000
	(.059)	(.066)	(.149)
Men x Age x Financial and Insurance Activities	.117*	.046	.346**
	(.061)	(.067)	(.144)
Men x Age x Professional, Scientific and Technical Activities	.048	.067	.429***
	(.057)	(.044)	(.072)
Men x Age x Administrative and Support Service	056	112**	.206***
Mon x Ago x Public Administration and Defence	(.071)	(.054)	(.076)
Men x Age x I ubic Administration and Defence	059 (063)	(076)	(110)
Men x Age x Education	014	.008	.363***
5	(.050)	(.047)	(.110)
Men x Age x Human Health and Social Work	033	.047	. 003
	(.087)	(.051)	(.157)
Men x Age x Arts, Entertainment, Recreation and Other	.299	.050	.338***
NT.	(.254)	(.050)	(.079)
IN	4954		

Notes: The dependent variable is employment status in April/May 2020. The model is estimated on the sample of employees, aged 20-63, with positive earnings from employment only in January/February 2020. The base outcome is no drop in earnings, i.e. of no more than £5 per week. The outcome reduced hours and earnings implies a fall of more than £5 in earnings and 1 working hour per week. Standard errors at a confidence level of 95% are shown in parenthesis. Significance levels indicated as * p < 0.1, ** p < 0.05, *** p < 0.01. Source: Own calculations with Understanding Society COVID-19 data.

	Reduced hours & earnings	Unemployed
Constant	-1.059	-3.782**
	(.830)	(1.899)
Men	280	-1.575
	(.438)	(1.059)
Age	004 (011)	.006
Household type:	(.011)	(.020)
Single adult, no children	.233	655
0 /	(.331)	(.758)
Single adult living with children	.229	562
	(.332)	(.756)
Multiple adults, no children	.448	-1.112
	(.325)	(.815)
Multiple adults living with children	.332	.475
Formin as ventile.	(.384)	(.797)
Larnings ventue:	rof	rof
1 9	- 481	169
2	(481)	$(1\ 052)$
3	1 254**	1 516
-	(.519)	(1.235)
4	.416	241
	(.521)	(1.464)
5	.647	1.506
	(.637)	(1.618)
6	057	-14.304***
	(.668)	(1.218)
7	.528	2.311^{*}
	(.572)	(1.358)
8	1.256	-13.852***
0	(1.157)	(1.610)
9	.762	-14.031^{++++}
10	(.790) 1 296**	(1.279)
10	(580)	(1.654)
11	1 289*	1 948
	(.682)	(1.319)
12	.287	-13.840***
	(.636)	(1.234)
13	.384	-14.126***
	(.724)	(1.435)
14	.283	-13.641^{***}
	(.545)	(1.229)
15	.082	1.434
10	(.579)	(1.801)
16	1.159**	2.689^{+}
17	(.023)	(1.041) 19 999***
11	.410	(1 494)
18	(.085)	(1.424) 3 211**
10	(.656)	(1.499)
19	.418	1.540
	(.562)	(1.687)
20	335	1.818
	(.524)	(1.352)
Sex x Hours in work:		
Women x less than 20	ref	ref
Men x less than 20	341	658
Wesser 20.24	(.512)	(1.279)
women x 20-34	.233	-1.009
Mon x 20.34	(.432)	(.991)
WICH X 20-04	.460 (277)	.922 (050)
Women x $35+$	(.577)	(. <i>3</i> 50) _ 040
	(.448)	(1.143)
Sector:	(.110)	(1110)
Agriculture, Mining, Manufacture, Utilities	\mathbf{ref}	ref
Construction and Real Estate	1.933***	2.531^{**}

Table A.2: Multinomial logit of self-employment status in
 \underline{April} 2020
(base outcome = no drop in earnings)

Wholesale and Retail Trade 1.291^{***} .611 (.472) (1.475) Transportation and Storage 1.347^{**} -2.821^{**} (.537) (1.363) Accommodation and Food 3.787^{***} 4.080^{**} (1.144) (1.644) Information and Communication 1.205^{**} 2.892^{***} (.577) (1.033) Financial and Insurance Activities 589 -15.483^{***} (.959) (1.138) Professional, Scientific and Technical Activities $.324$ 539 (.455) (1.392) Administrative and Support Service 1.729^{***} 3.336^{***} (.670) (1.078) Education .654 .548 (.452) (1.188) Human Health and Social Work .830* 857 Arts, Entertainment, Recreation and Other 1.676^{***} 1.166 (.472) (1.126) N		(.478)	(1.033)
	Wholesale and Retail Trade	1.291***	.611
$\begin{array}{llllllllllllllllllllllllllllllllllll$		(.472)	(1.475)
$ \begin{array}{cccc} (.537) & (1.363) \\ Accommodation and Food & 3.787^{**} & 4.080^{**} \\ (1.144) & (1.644) \\ Information and Communication & 1.205^{**} & 2.892^{***} \\ (1.144) & (1.644) \\ Information and Communication & 1.205^{**} & 2.892^{***} \\ (.577) & (1.033) \\ Financial and Insurance Activities & .589 & -15.483^{***} \\ (.959) & (1.138) \\ Professional, Scientific and Technical Activities & .324 &539 \\ (.455) & (1.392) \\ Administrative and Support Service & 1.729^{***} & 3.336^{***} \\ (.537) & (1.067) \\ Public Administration and Defence & (.537) & (1.067) \\ Public Administration and Defence & (.670) & (1.078) \\ Education & .654 & .548 \\ Human Health and Social Work & .830^* &857 \\ (.445) & (1.518) \\ Arts, Entertainment, Recreation and Other & 1.676^{***} & 1.166 \\ (.472) & (1.126) \\ N & 1365 \\ \end{array} $	Transportation and Storage	1.347**	-2.821**
Accommodation and Food 3.787^{***} 4.080^{**} Information and Communication 1.205^{**} 2.892^{***} Information and Insurance Activities 589 -15.483^{***} Information and Insurance Activities $.324$ 539 Professional, Scientific and Technical Activities $.324$ 539 Information and Support Service 1.729^{***} 3.336^{***} Information and Defence $(.455)$ (1.078) Education $.654$ $.548$ Human Health and Social Work $.830^*$ 857 Arts, Entertainment, Recreation and Other 1.676^{***} 1.166 N 1365 1365		(.537)	(1.363)
$ \begin{array}{cccc} (1.144) & (1.644) \\ \text{Information and Communication} & 1.205^{**} & 2.892^{***} \\ & (.577) & (1.033) \\ \text{Financial and Insurance Activities} &589 & -15.483^{***} \\ & (.959) & (1.138) \\ \text{Professional, Scientific and Technical Activities} & .324 &539 \\ & (.455) & (1.392) \\ \text{Administrative and Support Service} & 1.729^{***} & 3.336^{***} \\ & (.537) & (1.067) \\ \text{Public Administration and Defence} & 1.126^{*} & -14.502^{***} \\ & (.670) & (1.078) \\ \text{Education} & .654 & .548 \\ & (.452) & (1.188) \\ \text{Human Health and Social Work} & .830^{*} &857 \\ & (.445) & (1.518) \\ \text{Arts, Entertainment, Recreation and Other} & 1.676^{***} & 1.166 \\ & (.472) & (1.126) \\ \text{N} & 1365 \\ \end{array} $	Accommodation and Food	3.787^{***}	4.080**
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		(1.144)	(1.644)
$ \begin{array}{cccc} (.577) & (1.033) \\ \hline & \mbox{Financial and Insurance Activities} &589 &15.483^{***} \\ & (.959) & (1.138) \\ \hline & \mbox{Professional, Scientific and Technical Activities} & .324 &539 \\ & (.455) & (1.392) \\ \hline & \mbox{Administrative and Support Service} & 1.729^{***} & 3.336^{***} \\ & (.537) & (1.067) \\ \hline & \mbox{Public Administration and Defence} & 1.126^* &14.502^{***} \\ & (.670) & (1.078) \\ \hline & \mbox{Education} & .654 &548 \\ & (.452) & (1.188) \\ \hline & \mbox{Human Health and Social Work} & .830^* &857 \\ & (.445) & (1.518) \\ \hline & \mbox{Arts, Entertainment, Recreation and Other} & 1.676^{***} & 1.166 \\ & (.472) & (1.126) \\ \hline & \mbox{N} & 1365 \\ \end{array} $	Information and Communication	1.205^{**}	2.892***
		(.577)	(1.033)
$ \begin{array}{ccc} (.959) & (1.138) \\ (.455) & .324 & .539 \\ (.455) & (1.392) \\ (.455) & (1.392) \\ (.455) & (1.392) \\ (.455) & (1.392) \\ (.455) & (1.392) \\ (.455) & (1.67) \\ (.537) & (1.067) \\ (.537) & (1.067) \\ (.670) & (1.078) \\ $	Financial and Insurance Activities	589	-15.483^{***}
$\begin{array}{cccc} \mbox{Professional, Scientific and Technical Activities} & .324 &539 \\ & (.455) & (1.392) \\ \mbox{Administrative and Support Service} & 1.729^{***} & 3.336^{***} \\ & (.537) & (1.067) \\ \mbox{Public Administration and Defence} & 1.126^{*} & -14.502^{***} \\ & (.670) & (1.078) \\ \mbox{Education} & .654 & .548 \\ & (.452) & (1.188) \\ \mbox{Human Health and Social Work} & .830^{*} &857 \\ & (.445) & (1.518) \\ \mbox{Arts, Entertainment, Recreation and Other} & 1.676^{***} & 1.166 \\ & (.472) & (1.126) \\ \mbox{N} & 1365 \\ \end{array}$		(.959)	(1.138)
$\begin{array}{cccc} (.455) & (1.392) \\ \text{Administrative and Support Service} & 1.729^{***} & 3.336^{***} \\ (.537) & (1.067) \\ \text{Public Administration and Defence} & 1.126^{*} & -14.502^{***} \\ (.670) & (1.078) \\ \text{Education} & .654 & .548 \\ (.452) & (1.188) \\ \text{Human Health and Social Work} & .830^{*} &857 \\ (.445) & (1.518) \\ \text{Arts, Entertainment, Recreation and Other} & 1.676^{***} & 1.166 \\ (.472) & (1.126) \\ \text{N} & 1365 \end{array}$	Professional, Scientific and Technical Activities	.324	539
$\begin{array}{llllllllllllllllllllllllllllllllllll$		(.455)	(1.392)
$ \begin{array}{ccc} (.537) & (1.067) \\ \text{Public Administration and Defence} & 1.126^* & -14.502^{***} \\ (.670) & (1.078) \\ \text{Education} & .654 & .548 \\ (.452) & (1.188) \\ \text{Human Health and Social Work} & .830^* &857 \\ (.445) & (1.518) \\ \text{Arts, Entertainment, Recreation and Other} & 1.676^{***} & 1.166 \\ (.472) & (1.126) \\ \text{N} & 1365 \\ \end{array} $	Administrative and Support Service	1.729^{***}	3.336^{***}
$\begin{array}{cccc} \mbox{Public Administration and Defence} & 1.126^* & -14.502^{***} \\ & (.670) & (1.078) \\ \mbox{Education} & .654 & .548 \\ & (.452) & (1.188) \\ \mbox{Human Health and Social Work} & .830^* &857 \\ & (.445) & (1.518) \\ \mbox{Arts, Entertainment, Recreation and Other} & 1.676^{***} & 1.166 \\ & (.472) & (1.126) \\ \mbox{N} & 1365 \end{array}$		(.537)	(1.067)
$ \begin{array}{cccc} (.670) & (1.078) \\ \hline \text{Education} & .654 & .548 \\ (.452) & (1.188) \\ \hline \text{Human Health and Social Work} & .830^* &857 \\ (.445) & (1.518) \\ \hline \text{Arts, Entertainment, Recreation and Other} & 1.676^{***} & 1.166 \\ (.472) & (1.126) \\ \hline \text{N} & 1365 \\ \end{array} $	Public Administration and Defence	1.126^{*}	-14.502^{***}
Education .654 .548 (.452) (1.188) Human Health and Social Work .830* 857 (.445) (1.518) Arts, Entertainment, Recreation and Other 1.676*** 1.166 (.472) (1.126) N 1365		(.670)	(1.078)
$ \begin{array}{ccc} (.452) & (1.188) \\ \mbox{Human Health and Social Work} & .830^{*} &857 \\ & (.445) & (1.518) \\ \mbox{Arts, Entertainment, Recreation and Other} & 1.676^{***} & 1.166 \\ & (.472) & (1.126) \\ \mbox{N} & 1365 \\ \end{array} $	Education	.654	.548
Human Health and Social Work .830* 857 Arts, Entertainment, Recreation and Other (.445) (1.518) Arts, Entertainment, Recreation and Other 1.676*** 1.166 (.472) (1.126) N 1365		(.452)	(1.188)
	Human Health and Social Work	.830*	857
Arts, Entertainment, Recreation and Other 1.676*** 1.166 (.472) (1.126) N 1365		(.445)	(1.518)
N $(.472)$ (1.126) N 1365	Arts, Entertainment, Recreation and Other	1.676^{***}	1.166
N 1365		(.472)	(1.126)
	N	1365	

(table continued) Multinomial logit of self-employment status in \underline{May} 2020 (base outcome = no drop in earnings)

	Reduced hours	Unemployed
	& earnings	
Constant	765	-17.133***
	(1.028)	(2.465)
Men	625	-2.465*
	(.474)	(1.353)
Age	012	.017
	(.012)	(.037)
Household type:		
Single adult, no children	214	479
	(.366)	(.714)
Single adult living with children	.167	-1.692
	(.356)	(1.325)
Multiple adults, no children	192	-1.798*
	(.397)	(1.012)
Multiple adults living with children	681	656
	(.426)	(.737)
Earnings ventile:		
1	ref	ref
2	.009	180
	(.505)	(1.057)
3	.463	525
	(.541)	(1.586)
4	.681	1.559
	(.545)	(1.326)
5	.880	-13.742***
	(.658)	(1.390)
6	.062	-15.467^{***}
	(.657)	(1.453)
7	.276	3.012^{*}
	(.904)	(1.602)
8	.804	1.299
	(.715)	(1.629)
9	.727	-14.601***
	(.831)	(1.086)
10	1.140^{**}	.340
	(.577)	(1.517)
11	1.227	1.477
	(.800)	(1.962)
12	1.337^{*}	-13.446***
	(.706)	(1.666)

	(.870)	(1.653)
14	.807	-13.703***
	(.584)	(1.464)
15	1.539^{**}	1.722
	(.732)	(1.903)
16	1.791*	-13.965***
	(1.048)	(1.972)
17	2.254***	1.378
10	(.676)	(2.086)
18	.402	-15.490^{++++}
10	(.124) 9.690***	(1.292) 2.41.4*
19	2.020	(2.024)
20	(.000) 1.226*	(2.024) 3.931*
20	(633)	(1.716)
Sex x Hours in work	(.000)	(1.110)
Women x less than 20	ref	ref
Men x less than 20	.039	3.100**
	(.584)	(1.213)
Women x 20-34	.318	-1.177
	(.471)	(1.023)
Men x 20-34	1.123^{***}	1.343
	(.404)	(.836)
Women x 35+	.142	-2.240*
	(.456)	(1.237)
Sector:		
Agriculture, Mining, Manufacture, Utilities	ref	ref
Construction and Real Estate	.598	16.229^{***}
	(.636)	(1.017)
Wholesale and Retail Trade	.827	15.412***
m	(.653)	(1.191)
Transportation and Storage	.5((15.018^{++++}
Accommodation and Food	(./10) 2.081**	(1.104) 16 001***
Accommodation and Food	(811)	(1,434)
Information and Communication	(.011)	15 146***
mormation and communication	(733)	(1 149)
Financial and Insurance Activities	-1.661	-2.864**
	(1.117)	(1.367)
Professional, Scientific and Technical Activities	.460	13.804***
· · · · · · · · · · · · · · · · · · ·	(.647)	(1.505)
Administrative and Support Service	1.088	15.957***
	(.709)	(.907)
Public Administration and Defence	.165	-1.335
	(.738)	(1.066)
Education	.495	667
	(.638)	(1.201)
Human Health and Social Work	.676	13.987***
	(.633)	(1.388)
Arts, Entertainment, Recreation and Other	1.307**	14.593***
NY.	(.650)	(1.031)
N	1074	

Notes: The dependent variable is self-employment status in April/May 2020. The model is estimated on the sample of self-employed, aged 20-63, with positive earnings from self-employment in January/February 2020. The base outcome is no drop in earnings, i.e. of no more than £5 per week. The outcome reduced hours and earnings implies a fall of more than £5 in earnings and 1 working hour per week. Significance levels indicated as * p < 0.1, ** p < 0.05, *** p < 0.01.

Source: Own calculations with Understanding Society COVID-19 data.

Table A.3:	Logit for	receiving th	ne SEISS	grant in	May 2	2020

Constant	-1.633
	(1.031)
Men	1.153**
	(.446)
Age	.005
	(.012)
Self-employment status in May vs Jan/Feb:	. ,
No change	ref
With reduced hours and earnings	.420*
•	

The second second	(.229)
Unemployed	(.516)
Household type:	、 , ,
Single adult, no children	.393
Single adult living with children	.309
Multiple adults, no children	(.464) - 262
Multiple adults, no children	(.512)
Multiple adults living with children	.360
Earnings ventile:	(.020)
1	ref 286
2	(.562)
3	083
4	.051
F	(.567)
5	(.734)
6	069
7	(.670) 483
	(.799)
8	074 (.710)
9	-1.192
10	(1.028) .239
	(.608)
11	-1.009
12	-1.122
13	(.768)
19	(1.073)
14	297
15	781
16	(.681)
10	(.960)
17	959
18	(.696) 687
10	(.771)
19	398 (.600)
20	-1.446**
Sex x Hours in work:	(.600)
Women x less than 20	ref
Men x less than 20	-2.613^{***} (.621)
Women x 20-34	.678
Men x 20-34	(.461) 476
	(.385)
Women x 35+	.603 (.447)
Sector:	()
Agriculture, Mining, Manufacture, Utilities Construction and Real Estate	ref .838
	(.580)
Wholesale and Retail Trade	.417
Transportation and Storage	1.609**
Accommodation and Food	(.698) 1 223
	(.913)
Information and Communication	.370

	(.613)
Financial and Insurance Activities	.008
	(1.011)
Professional, Scientific and Technical Activities	.475
	(.555)
Administrative and Support Service	1.053^{*}
	(.580)
Public Administration and Defence	252
	(.828)
Education	.443
	(.582)
Human Health and Social Work	.422
	(.550)
Arts, Entertainment, Recreation and Other	1.255^{**}
	(.578)
Ν	1074

Notes: The model is estimated on the sample of self-employed, aged 20-63, with positive earnings from self-employment in January/February 2020. Standard errors at a confidence level of 95% are shown in parenthesis. Significance levels indicated as * p < 0.1, ** p < 0.05, *** p < 0.01. Source: Own calculations with Understanding Society COVID-19 data.

	no earner	1 earner	2+ earners	with	with	with	with	lone mother
				children	3+ children	disability	elderly	
decile								
1	22.3	9.7	1.6	9.6	18.4	6.9	12.0	15.2
2	15.7	12.2	2.5	12.6	22.6	11.2	12.1	21.8
3	14.6	11.9	4.5	12.2	17.4	13.9	13.3	20.3
4	12.8	10.6	7.0	11.9	11.4	19.5	12.4	14.0
5	10.2	11.2	8.7	10.9	8.3	17.3	11.7	10.5
6	8.1	9.7	11.6	10.4	5.5	13.9	10.6	7.2
7	5.2	9.4	14.1	9.6	4.7	8.1	8.2	6.2
8	4.1	8.1	15.7	8.3	4.5	4.5	7.3	1.9
9	3.7	8.3	16.6	7.0	3.1	3.0	6.3	1.9
10	3.3	8.7	17.8	7.4	4.1	1.6	6.1	.9
n households	9,826,848	7,949,500	10,049,547	7,801,746	1,292,746	$3,\!679,\!514$	8,597,443	1,661,870

 Table A.4: Household composition in the income distribution

(table continued)

	single-person hh:	single-person hh:		accommodation		
	women	men	own	private rent	social rent/other	
decile						
1	18.9	17.6	9.5	10.2	18.4	
2	14.0	10.3	7.9	10.5	16.7	
3	13.3	12.2	7.2	11.3	19.6	
4	13.4	10.6	8.2	11.5	15.1	
5	10.8	9.9	9.5	10.5	11.2	
6	9.6	8.2	10.1	10.2	8.5	
7	6.7	7.7	11.0	9.3	5.2	
8	4.8	6.5	11.3	9.4	2.9	
9	3.9	8.5	12.1	9.1	1.7	
10	4.6	8.7	13.3	8.0	.8	
n households	4,308,600	3,795,195	17,687,422	5,112,757	5,025,716	

Notes: The table shows, for a given household type, what proportion (in %) of households are situated in each decile group. Income decile groups based on household equivalised net income before Covid-19. Each column refers to a household with certain characteristics. Household types for presence of earners are based on employment status before Covid-19. Household types can be overlapping.

Source: Own calculations with UKMOD and FRS.

	Baseline	Impact of crisis	
	(in %)	(change t	o baseline)
		fixed	floating
All	16.491***	1.173***	-2.744***
	(.350)	(.226)	(.257)
Women	17.040***	1.161^{***}	-2.873***
	(.397)	(.215)	(.268)
Men	15.927^{***}	1.185^{***}	-2.612***
	(.388)	(.268)	(.293)
Children	21.505^{***}	1.514^{***}	-4.484***
	(.715)	(.461)	(.491)

Table A.5: Change in the poverty rate in % points

Notes: The table shows the impact on the poverty rate of the employment and earnings shocks and the UK policy response, i.e. the *baseline* versus scenario D. The poverty rate is the % of people with household equivalised net income below the poverty line. The poverty line is 60% of the median household equivalised net income in the baseline (fixed) or of the respective scenario (floating). Standard errors at a confidence level of 95% are shown in parenthesis. Bootstrapped standard errors after 200 replications. Significance levels indicated as * p < 0.1, ** p < 0.05, *** p < 0.01.

Source: Own calculations with UKMOD and FRS.

	Baseline (in %)	Impact of crisis (change to baseline)		
		fixed	floating	
All	4.164***	.383***	452***	
	(.116)	(.043)	(.062)	
Women	4.194***	.392***	479***	
	(.120)	(.045)	(.065)	
Men	4.133***	.373***	425***	
	(.137)	(.047)	(.065)	
Children	4.980***	.646***	434***	

Table A.6: Change in the poverty gap in % points

Notes: The table shows the impact on the poverty gap of the employment and earnings shocks and the UK policy response, i.e. the *baseline* versus scenario D. The poverty gap is the mean shortfall of the total population from the poverty line (counting the non-poor as having zero shortfall), expressed in % of the poverty line. The poverty line is 60% of the median household equivalised net income in the baseline (fixed) or of the respective scenario (floating). Standard errors at a confidence level of 95% are shown in parenthesis. Bootstrapped standard errors after 200 replications. Significance levels indicated as * p < 0.1, ** p < 0.05, *** p < 0.01. Source: Own calculations with UKMOD and FRS.

(.091)

(.113)

(.244)

	Baseline	Impact of crisis (change to baseline)					
	(in levels)	E + K	S_t	S_b	Р	Δ	
Gini	.310***	.005***	.003***	013***	005***	010***	
	(.003)	(.002)	(.000)	(.000)	(.000)	(.001)	
CV	.685***	.003	.007	027***	009***	025*	
	(.029)	(.027)	(.010)	(.002)	(.000)	(.015)	
MLD	.165***	.004	.003	011***	005***	010***	
	(.004)	(.003)	(.002)	(.002)	(.000)	(.001)	
TI	.169***	.005	.002	012^{***}	005***	010***	
	(.006)	(.005)	(.002)	(.001)	(.000)	(.003)	

Table A.7: Decomposing changes in income inequality

Notes: Inequality estimates based on equivalised household income. The total change Δ is decomposed into: the contribution of earnings changes (E + K), tax/NIC and benefit automatic stabilisers $(S_t \text{ and } S_b)$, and Covid-related benefit increases (P). Standard errors at a confidence level of 95% are shown in parenthesis. Bootstrapped standard errors after 200 replications. Significance levels indicated as * p < 0.1, ** p < 0.05, *** p < 0.01. Source: Own calculations with UKMOD and FRS.

B Figures





Notes: The figure shows the distributional impact of the employment and earnings shocks and the UK policy response, i.e. the *baseline* versus scenario D. Changes in income based on equivalised household net income. All-population deciles.

Source: Own calculations using UKMOD and FRS.



Figure B.2: Impact of the crisis and policy response to Covid-19 on mean net income by decile: households with 2+ earners

Notes: The figure shows the distributional impact of the employment and earnings shocks and the UK policy response, i.e. the *baseline* versus scenario *D*. Changes in income based on equivalised household net income. All-population deciles.

Source: Own calculations using UKMOD and FRS.



Figure B.3: Impact of the crisis and policy response to Covid-19 on mean net income by decile and tax-benefit policy: households with 1 earner

Notes: Left plot shows impact of automatic stabilisers, i.e. the baseline versus scenario C. Right plot shows impact of Covid-related benefit increases, i.e. scenario C versus D. Changes in total net income and the contribution of earnings changes, CJRS subsidies and SEISS grant are omitted. Changes in income based on equivalised household net income. Other benefits include the Council Tax Reduction, Child Tax Credit, Income support, income-related Employment and Support Allowance, income-based JSA, Pension Credit, Scottish benefits (Sure Start Maternity Grant and Best Start Grant). No simulations to Statutory Sickness Pay. Source: Own calculations using UKMOD and FRS.



Figure B.4: Impact of the crisis and policy response to Covid-19 on mean net income by decile and tax-benefit policy: households with 2+ earners

Notes: Left plot shows impact of automatic stabilisers, i.e. the baseline versus scenario C. Right plot shows impact of Covid-related benefit increases, i.e. scenario C versus D. Changes in total net income and the contribution of earnings changes, CJRS subsidies and SEISS grant are omitted. Changes in income based on equivalised household net income. Other benefits include the Council Tax Reduction, Child Tax Credit, Income support, income-related Employment and Support Allowance, income-based JSA, Pension Credit, Scottish benefits (Sure Start Maternity Grant and Best Start Grant). No simulations to Statutory Sickness Pay. Source: Own calculations using UKMOD and FRS.



Figure B.5: Impact of the crisis and policy response to Covid-19 on mean net income by household type

Notes: The figure shows the distributional impact of the employment and earnings shocks and the UK policy response, i.e. the *baseline* versus scenario *D*. Changes in income based on equivalised household net income. Household types for presence of earners are based on employment status before Covid-19. *Source:* Own calculations using UKMOD and FRS.





Notes: Left plot shows impact of automatic stabilisers, i.e. the baseline versus scenario C. Right plot shows impact of Covid-related benefit increases, i.e. scenario C versus D. Changes in total net income and the contribution of earnings changes, CJRS subsidies and SEISS grant are omitted. Changes in income based on equivalised household net income. Household types for presence of earners are based on employment status before Covid-19. Other benefits include the Council Tax Reduction, Child Tax Credit, Income support, income-related Employment and Support Allowance, income-based JSA, Pension Credit, Scottish benefits (Sure Start Maternity Grant and Best Start Grant). No simulations to Statutory Sickness Pay.

Source: Own calculations using UKMOD and FRS.



Figure B.7: Impact of the crisis and policy response to Covid-19 on mean net income by age group

Notes: The figure shows the distributional impact of the employment and earnings shocks and the UK policy response, i.e. the *baseline* versus scenario *D*. Changes in income based on equivalised household net income. *Source:* Own calculations using UKMOD and FRS.

C Benefit take-up

The calculations of means-tested benefits and tax credits in UKMOD account for non-take-up. Take-up rates are based on the 2017 mid-point estimates on a caseload basis by the Department for Work and Pensions and HM Revenue and Customs.¹⁰ Households from the FRS sample are randomly selected to take-up their simulated entitlements so that the number of takers is in line with the official take-up rates. For UC, there is no evidence yet for the extent of non-take-up. However, due to the means-tested nature of the benefit, it is plausible that, like the benefits UC is replacing, UC does not reach all entitled families. A take-up rate of 87% is assumed

¹⁰DWP 2017/18 take-up estimates for income-related benefits available here: https://www.gov.uk/government/statistics/income-related-benefits-estimates-of-take-up-financial-year-2017-to-2018. HMRC 2017/18 take-up estimates for tax credits available here: https://www.gov.uk/government/statistics/child-benefit-child-tax-credit-ctc-and-working-tax-credit-wtc-take-up-rates-2017-to-2018

for UC (as the take-up rate estimated for the Income Support benefit for families without children). For more information on the take-up assumptions in UKMOD, see Reis and Tasseva (2020).

In scenarios C and D, we assume that families affected by the shocks do not take up UC if they were entitled to and did not take up UC prior to the shock. That is their take-up behaviour remains the same even after the change in their circumstances. In the case of affected families who become newly entitled to UC after the shock, the majority takes up UC but a small proportion of eligible families (as with Income Support) do not receive UC (e.g. due to errors in assessing their eligibility).